July 27, 2007

Optical diagnostic: CERN agenda (assuming no chance to pull snout out for alignment)
1. Prior to tunnel access, confirm all camera controls are live and good in the control room. Run/trigger all cameras. Leave live images on all monitors prior entrance to tunnel.

2. Upon enter the tunnel. Fix Olympus camera problem. Stop laser trigger. Send red light to confirm fiber continuity. Check back connector on the back of diode laser, and visually examine whether camera mount got displaced. Better secure them otherwise.
3. Pull Wind98 machine out and re-sit frame grabber board, start SMD camera to see if viewport #2 still have image collection problem (you don’t need laser light to do this). Otherwise remove frame grabber boards (two), SMD camera, power supply for the SMD camera, and all its associated cables out of the tunnel. Bring them back to US. Put Win98 machine back on the rack with IEEE connection.
4. Assume that Olympus camera got fixed. Then move imaging fiber #2 to Olympus camera and move illumination (Orange) fiber #2 to Laser diode output #4.. i.e. switch Viewport #2 to Viewport #4.

5. Clean up and organize ethernet cables/wires/fibers near the optics rack.
6. Confirm image collected on all cameras locally in TT2A and then in the control room including remote power recycles. 

